Author's response to reviews

Title: Gender distribution of adult patients on highly active antiretroviral therapy (HAART) in Southern Africa: a systematic review

Authors:

Adamson S Muula (muula@email.unc.edu)
Thabale J Ngulube (thabalejackngulube@gmail.com)
Seter Siziya (ssiziya@yahoo.com)
Cecilia M Makupe (cmakupe@chanco.unima.mw)
Eric Umar (ericumar@yahoo.com)
Hans W Prozesky (hwp@sun.ac.za)
Charles S Wiysonge (wiysongeC@unaids.org)
Ronald H Mataya (rmataya@llu.edu)

Version: 7 Date: 24 April 2007

Author's response to reviews: see over
23 April 2007

Submission of revision

The Editor’s comments are as below
2. The table on page fourteen extending to page fifteen needs to be centred on the page,
3. Page nine could the word 'manuscript' be taken back to the previous page and a full point added.
4. Page two: could the line spacing between the conclusion and Background paragraphs be adjusted. I think removing one line space should be fine.

Our responses.
All these suggested changes have been made. The website is now reference 13.
We have effected the other changes. Thank you.

Adamson Muula